Author’s response to reviews

Title: Toothache and associated factors in Brazilian adults: a cross-sectional population-based study.

Authors:

Mirian Kuhnen (kuhnen@uniplac.net)
Marco A Peres (mperes@ccs.ufsc.br)
Anelise V Masiero (anemasiero@uniplac.net)
Karen G Peres (karengp@ccs.ufsc.br)

Version: 13 Date: 22 January 2009

Author's response to reviews: see over
Dr. Rikki Graham  
Senior Assistant Editor  
BMC-series journals

Dear Dr. Graham,

On behalf of all authors, I am submitting the final version with another correction of the manuscript “Toothache and associated factors in Brazilian adults: a cross-sectional population-based study” to the BMC Oral Health.

The text has been revised according to minor pending formatting. The changes made were listed.

Yours sincerely,

Karen Glazer Peres  
Federal University of Santa Catarina, Brazil.
Minor revisions

Table 1: It was added the number of individuals in the variables: *Per capita income*-BMW, Educational attainment-years, and Consumption of cigarettes- Package year.

Table 2: we have changed the information level 3 variables from Model 2 to Model 4 again, in order to correct the column position.